Author’s response to reviews

Title: Bilateral Spontaneous Lateral Intrasphenoidal and Transethmoidal Meningoceles: A rare presentation of Idiopathic Intracranial Hypertension

Authors:

Fahad Alkherayf (falkherayf@toh.ca)
Abdul Mounem Kassab (ayass056@uottawa.ca)
Aleksandar Radonjic (arado058@uottawa.ca)
Ioana Moldovan (imoldovan@ohri.ca)
Shaun Kilty (skilty@toh.ca)

Version: 2 Date: 20 Nov 2018

Author’s response to reviews:

Reviewer reports:

Give details about complete physical and neurological examination. Give pulse, BP on admission. Give results of laboratory tests: blood count, Chemistry, liver function and CSF analysis. Add paragraph at the beginning of the Discussion to summarize the case.

FA: complete physical and neurological examination including pulse and BP on admission, as well as all available laboratory tests results were added to the "Case presentation" section of the manuscript. Also, a paragraph to summarize the case was added at the beginning of the "Discussion" section of the manuscript.